Author’s response to reviews

Title: Pulmonary arteriovenous malformations presenting as difficult-to-control asthma: a case report

Authors:

Marta Navratil MD (marta.navratil@gmail.com)
Vinko Vidjak MD, PhD, Assistant Professor (vinko.vidjak@gmail.com)
Filip Rubić MD (filiprubic@gmail.com)
Damir Erceg MD (erceg@bolnica-srebrnjak.hr)
Mirjana Turkalj MD, PhD (turkalj@bolnica-srebrnjak.hr)

Version: 2 Date: 11 August 2012

Author’s response to reviews:

1. The Abstract has been reconstructed into the following three sections: Introduction, Case presentation, and Conclusion.
2. The header “Background” has been changed to “Introduction.”
3. We used the term male in describing the patient’s gender.
4. We included a competing interests section at the end of the manuscript.
5. We added the last sentence of the author’s contribution section: ‘All authors read and approved the final manuscript.’
6. Signed consent form will be send by email.